Author’s response to reviews

Title: Medical Care Usage and Self-Rated Mental Health

Authors:

Prof James E Rohrer Ph.D. (james.rohrer@ttuhsc.edu)

Version: 4 Date: 16 Feb 2004

1. In the next-to-the-last paragraph of the Results section, "ten or more" has been changed to "more than ten".
2. 'Multivariate' has been added to the title of the table.
3. The baseline category has been added to the table with odds ratio shown as 1.00
4. A physical health variable was not added because the available variable is overall self-rated health. This obviously encompasses mental health, which confuses the model. Since this was a discretionary revision, not acting on it should not pose any concerns.